Author's response to reviews

Title: A phase II, open-label, multicentre study to evaluate the immunogenicity and safety of an adjuvanted pre-pandemic (H5N1) influenza vaccine in healthy Japanese adults

Authors:

- Hideaki Nagai (hnagai-in@tokyo-hosp.jp)
- Hideyuki Ikematsu (ikematsu@gray.plala.or.jp)
- Kazuyoshi Tenjinbaru (kazyoshi.2.tenjinbaru@gsk.com)
- Atsushi Maeda (atsushi.2.maeda@gsk.com)
- Mamadou Dramé (mamadou.drame@gsk.com)
- François P Roman (francois.p.roman@gsk.com)

Version: 4 Date: 23 November 2010

Author's response to reviews: see over
Ref: 1452825006411612: “A phase II, open-label, multicentre study to evaluate the immunogenicity and safety of an adjuvanted prepandemic (H5N1) influenza vaccine in healthy Japanese adults” by Hideaki Nagai, Hideyuki Ikematsu, Kazuyoshi Tenjinbaru, Atsushi Maeda, Mamadou Dramé and François P Roman.

Dear Dr. Norton,

Thank you for your email. As requested, we changed the numerical order of the references and we provided a functioning web link for reference 23.

Thank you for your help.

Yours sincerely,

Dr. Hideaki Nagai
Corresponding author
Address: National Hospital Organization Tokyo National Hospital, 3-1-1, Takeoka, Kiyose-city, Tokyo 204-8585, Japan; Email: hnagai-in@tokyo-hosp.jp